
Daily Status Tues 16-Oct-12
16-Oct-12

Personnel: JohnM.,  TomH.

Weather&Ops:
Remained mostly clear last night with calm/light wind.    Clear sky in am, warmish seasonal temps.  Winds expected to pick up today and cooler tomorrow 
with wind.   A squall with rain and frontal passage came through at ~20MST in the evening, 2Z on 17th.

Sensors:
Overall things looking good / reporting.   Nearly everything is reporting except the Picarro, which is still awaiting to be repaired at their factory.  

Kryptons: voltages a bit better this morning, but we expect to clean them today.....  Kryptons were cleaned at roughly 10MDT, 16Z.   The voltages did 
come up a bit to roughly 1.2.   

TRH's maintenance and weatherproofing of binder connectors will continue today causing outages.....see specific log entry table.
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